Author's response to reviews

Title: Exploring the Usability of a Videophone Mock-up for People with Dementia and their Significant Others

Authors:

Inga-Lill Boman (inga-lill.boman@ds.se)
Stefan Lundberg (stefan.lundberg@chb.kth.se)
Sofia Starkhammar (sofia.starkhammar@sll.se)
Louise Nygård (louise.nygard@ki.se)

Version: 2
Date: 9 December 2013

Author's response to reviews:

Dear Catherine Olino,

Thank you very much for the instructions how to make the changes.

1. A title page have been included at the front of the manuscript file. It contains the title, author details and full address of the submitting author.
2. A section with ethics and considerations have been included.
3. A figure title have been included after the references.
4. Figure 2 have been removed. It should be table 2. Table 2 have been included after table 1 in the main manuscript.
5 a new main document have been uploaded